Author's response to reviews

Title: Design of a factorial experiment with randomization restrictions to assess medical device performance on vascular tissue.

Authors:

Wiebke S Diestelkamp (Wiebke.Diestelkamp@notes.udayton.edu)
Carissa M Krane (Carissa.Krane@notes.udayton.edu)
Margaret F Pinnell (Margaret.Pinnell@notes.udayton.edu)

Version: 2 Date: 17 May 2010

Author's response to reviews:

We originally submitted this manuscript for consideration for publication in BMC Bioinformatics. Upon the recommendation of Dr. Maria Kowalczuk, Deputy Biology Editor for BMC series Journals, we have reformatted the manuscript to conform with the style of BMC Medical Research Methodology as a Study Protocol. We have changed the abstract to contain sections with subheadings Background, Methods/Design, Discussion. We have changed the body of the text to include subsection headings Background, Methods/Design, and Discussion, and have included the requested Competing Interests Section at the end of the Manuscript.